Author's response to reviews

Title: A randomized controlled trial of mental health interventions for survivors of systematic violence in Kurdistan, Northern Iraq

Authors:

Paul Bolton (pbolton1@jhu.edu)
Judith Bass (jbass1@jhu.edu)
Goran Zangana (gsabir@heartlandalliance.org)
Talar Kamal (talar_kamal@yahoo.com)
Sarah Murray (smmuray9@jhu.edu)
Debra Kaysen (dkaysen@uw.edu)
Carl Lejuez (clejuez@umd.edu)
Kristen Lindgren (kpl9716@uw.edu)
Sherry Pagoto (Sherry.Pagoto@umassmed.edu)
Laura Murray (lmurra15@jhu.edu)
Stephanie Skavenski (sskaven1@jhu.edu)
Nazar Amin (nazar.amin@gmail.com)
Ahmed Ahmed (ahmed.ammin@gmail.com)
Michael Rosenblum (mrosen@jhu.edu)

Version: 4 Date: 26 November 2014

Author's response to reviews:

our responses are included in the accompany cover letter.